

Network for Studies on Pensions, Aging and Retirement

To what extent can partial retirement ensure retirement income adequacy?

Tunga Kantarcı Jochem Zweerink

DESIGN PAPER 149

DESIGN PAPERS are part of the **refereed Industry Paper Series**, which are refereed by the Netspar Editorial Board. Design Papers discuss the design of a component of a pension system or product. A Netspar Design Paper analyzes the objective of a component and the possibilities for improving its efficacy. These papers are easily accessible for industry specialists who are responsible for designing the component being discussed. Authors are allowed to give their personal opinion in a separate section. Design Papers are presented for discussion at Netspar events. Representatives of academic and private sector partners, are invited to these events. Design Papers are published at the Netspar website.

Colophon Netspar Design Paper 149, May 2020

Editorial Board

Rob Alessie – University of Groningen Iwan van den Berg – AEGON Netherlands Mark-Jan Boes - VU Amsterdam Marijke Colly – MN Kees Goudswaard – Leiden University Arjen Hussem – PGGM Bert Kramer – University of Groningen & Ortec Finance Fieke van der Lecq (Chair) – VU Amsterdam Raymond Montizaan – Maastricht University Alwin Oerlemans – APG Maarten van Rooij – De Nederlandsche Bank Peter Schotman – Maastricht University Koen Vaassen – Achmea Mieke van Westing – Nationale Nederlanden Peter Wijn – APG Jeroen Wirschell - PGGM Marianne Zweers – a.s.r.

Design

B-more Design

Lay-out

Bladvulling, Tilburg

Editors

Frans Kooymans, Frans Kooymans-Text and Translation Netspar

Design Papers are publications by Netspar. No reproduction of any part of this publication may take place without permission of the authors.

CONTENTS

Su	mmary	4
Sa	menvatting	5
1.	Introduction	6
2.	The Dutch pension and income tax systems	8
3.	Data	13
4.	Method and assumptions	15
5.	Results	19
6.	Conclusion	34
Re	ferences	36
Ap	pendix	38

Acknowledgments

We wish to thank Erik ten Hove, Jan Maarten van Riemsdijk, and Joost van Valkengoed for supporting this research and providing helpful explanations of the institutional setting, and two anonymous reviewers for their insightful comments and suggestions. We thank PFZW for providing the data. This research is supported by the Network for Studies on Pensions, Aging and Retirement (Netspar) under grant number LMVP 2014.03. Its contents are the responsibility of the authors and do not necessarily represent the official views of Netspar.

Affiliations

Tunga Kantarcı – Tilburg University Jochem Zweerink – CPB

Summary

Policy makers are confronted with the challenge of making the pension income adequate for older individuals and the pension system sustainable at the same time. Partial retirement can be an instrument to achieve both goals. Combining wage and pension income can fund old-age consumption, and induce individuals to extend their working lives up to the legal retirement age or beyond. Partial retirement can facilitate employment especially among individuals with heavy occupations who struggle to work full-time until the legal retirement age. We analyze the implications of partial retirement for the financial well-being of older individuals, and to what extent this implies a self-reliant financial security in retirement. Using data on the pension entitlements of the clients of a large pension fund, we show that, on average, when individuals retire partially before the legal retirement age, they attain at least a minimum target replacement rate of 70 percent during partial retirement, and subsequently a higher replacement rate during full retirement. The replacement rates fall much below the minimum target replacement rate if individuals retire full-time. Furthermore, we show that when individuals retire part-time instead of full-time at a given age, they are much more likely to maintain an adequate pension income when pension funds do not index pension rights, or when pension income is smaller due to accruing pension rights for fewer years.

Samenvatting

Beleidsmakers staan voor de uitdaging om tegelijkertijd ouderen van een adequaat pensioeninkomen te voorzien en het pensioenstelsel houdbaar te houden. Deeltijdpensioen kan een instrument zijn om beide doelen te bereiken. De combinatie van loon- en pensioeninkomen kan consumptie op latere leeftijd financieren en individuen ertoe aanzetten hun werkende leven te verlengen tot de AOW-leeftijd of daarna. We analyseren de gevolgen van gedeeltelijke pensionering voor het financiële welzijn van ouderen en in hoeverre dit financiële zelfredzaamheid tijdens het pensioen inhoudt. We laten aan de hand van pensioenaansprakendata van klanten van een groot pensioenfonds zien dat individuen, gemiddeld genomen, wanneer zij met deeltijdpensioen gaan voor de AOW-leeftijd, de minimale streefvervangingsratio van 70% behalen tijdens het deeltijdpensioen en een nog hogere vervangingsratio tijdens voltijdpensioen. De vervangingsratio's liggen ver onder de minimale streefvervangingsratio wanneer zij voltijds met pensioen gaan. Verder laten we zien dat wanneer mensen op een bepaalde leeftijd deeltijds met pensioen gaan in plaats van voltijds, zij een hogere kans op een adequaat pensioeninkomen hebben wanneer pensioenfondsen pensioenrechten niet indexeren, of wanneer het pensioeninkomen lager is doordat zij minder jaren pensioenrechten op hebben opgebouwd.

1. Introduction

Recent policy measures in the Netherlands aim at tailoring individual retirement income to personal financial needs and at making the pension system financially sustainable. Employees bear increasing investment risk due to the indexation of their pension rights with respect to the returns on investment in the financial market. The accrual rate of occupational pension rights has been reduced in the past decade and may be reduced further in the coming years. The resulting reductions in accrued or paid pension rights are likely to make individuals more dependent on alternative sources of income during retirement or induce them to work longer to compensate for lost income. Furthermore, the pension law is expected to accelerate the shift from defined benefit to defined contribution schemes. Compared to a defined benefit scheme, in a defined contribution scheme the worker faces higher risks and greater personal responsibility to secure an adequate level of retirement income through decisions on contribution rates, asset allocation, or decumulation strategy. Such decisions require an adequate level of financial literacy (Poterba, 2014).

Partial retirement can help individuals to secure their financial well-being in retirement. For example, by working part-time at an otherwise early retirement age, individuals can rely on part-time labor earnings instead of only on an occupational pension that is actuarially penalized (Laczko, 1988; Kantarcı et al., 2013). While working part-time, they also continue to accrue pension rights for full retirement at a later age. Partial retirement can also be beneficial for public finances if it extends working lives. In fact, several studies have suggested that, in the United States and in Europe, supplementing retirement income with income from part-time work can ease the pressure on the pension system caused by population aging (Laczko, 1988; Chen, 1996; Cahill et al., 2006; Robinson and Clark, 2010). This is in line with the policy objective of the European Commission, which focuses on ensuring the adequacy of pensions in aging societies without straining public finances (European Commission, 2012).

Van Duijn et al. (2013) and Knoef et al. (2016) investigate retirement income adequacy, based on replacement rates when individuals retire full-time from their career job before or at the statutory retirement age. Kantarcı et al. (2013) compute replacement rates in full and partial retirement scenarios for a hypothetical worker with assumed job characteristics. These studies provide rich analyses on financial security in retirement as they compare actual and expected replacement rates, study the implications of partial retirement for replacement rates, or identify the vulnerable groups in the population that have replacement rates below a minimum target level or full replacement level. However, in these studies the calculation of the replacement rate is not based on actual wages and pension rights at the time of retirement, but on projections of wages and pension rights from the time of observation until the day of retirement, based on strong assumptions about wage profiles, work histories, and accumulation of pension rights.

We analyze the retirement income of clients of a large pension fund in relation to their pre-retirement earnings to investigate their financial well-being in retirement. As information on their accrued pension rights up to the retirement age is available, we make weak assumptions when analyzing pension income adequacy. We start by analyzing cases of full retirement, where pension plan participants stop working and claim full occupational pension rights at ages from 60 to 70. We then analyze cases where individuals partially retire and claim a fraction of their accrued pension rights at those same ages. We account for the receipt of state pension rights and the applicable income tax rules and tax credits. Our main aim is to show, among clients of a large pension fund, to what extent retiring part-time instead of full-time reduces the risk of falling below a minimum target replacement rate of 70% both before and after the statutory retirement age.

We show that, both before and after the statutory retirement age, workers attain a much higher level of retirement income security when they retire part-time and combine part-time earnings with a partial pension, compared to when they retire full-time and claim the full amount of their accrued pension rights. For example, if the sample of workers retire full-time at age 63, about 22% fall below the minimum target replacement rate at age 70, while if they continue working part-time from that same age for a period of five years, about 4% fall below the minimum target replacement rate at age 70. Furthermore, we show that when individuals retire parttime instead of full-time at a given age, they are much more likely to maintain an adequate pension income when pension funds do not index pension rights, or when pension income is smaller due to accruing pension rights for fewer years.

The remainder of this paper is organized as follows. Section 2 describes the institutional setting. Section 3 presents the data. Section 4 describes the assumptions and calculation of the replacement rates. Section 5 presents the results. Section 6 discusses policy implications and contains conclusions.

2. The Dutch pension and income tax systems

The Dutch retirement income stands on three pillars. The first pillar is the state pension, the second pillar is the occupational pension, and the third pillar is the individual savings. All retirees who never lived abroad receive the full state pension, and the majority participate in a mandatory occupational pension scheme that is of the defined benefit type. The share of the third pillar in retirement income is much smaller, and its importance varies much across individuals (Alessie and Kapteyn, 2001; Knoef et al., 2016).¹ Therefore, the role of the third pillar is not considered in this study. This fairly homogeneous pension system allows a systematic analysis of retirement income across a large population of retirees with otherwise heterogeneous characteristics.

The state pension scheme

The General Old–Age Pensions Act (AOW) is the state pension scheme, paying flatrate benefits to people above the statutory retirement age who have always lived in the Netherlands, independent of earnings, income or premiums paid (Sociale Verzekeringsbank, 2019).² It provides households with a subsistence level income in which the breadwinner attained the eligibility age or is older. The only thing that therefore matters for the benefit level is household composition: individuals who make up a couple receive less than single individuals, and there are special allow– ances for children younger than 21 and for single income couples in which only the breadwinner has reached the eligibility age.

The scheme is unfunded and based on the pay-as-you-go principle, meaning that current pensions are financed from the current premiums paid by workers. The premiums are paid as a percentage of work income through the personal income tax, and labeled as national insurance premiums. Every person who lives in the Netherlands is insured under the scheme. The maximum period of insurance is 50 years, from the age of 15 until the statutory retirement age. For those who do not live in the Netherlands during this entire time, the benefit is adjusted proportionally.

¹ Knoef et al. (2016) show that, in 2010, among people who were 60 to 64 years old, the median net replacement rate was 82% when state and occupational pension benefits are considered in the calculation, increasing to 92% when, in addition, voluntary third pillar pensions and private wealth are taken into account.

² The statutory retirement age is 66 years in 2018, and is being gradually increased to reach 67 in 2021. It will be partially linked to life expectancy as from 2022.

The state pension is paid as an annuity after the statutory retirement age, and calculated as the sum of the product of the accrual rate and pension base from each year over the period of insurance. The accrual rate is 2% per year. The pension base is determined by the government according to the net minimum wage. It depends on the domestic situation of the retiree: single without a child, single with a child under 18 years old, living with a partner who is older than 65 years, or living with a partner who is younger than 65 years, all with or without a supplementary allowance.

In the state pension scheme, it is not possible to claim all or part of one's accrued pension rights before or after the statutory retirement age. However, the government put forward proposals to introduce flexible retirement options into the state pension scheme in 2008 and more recently in 2019 (Ministerie van Sociale Zaken en Werkgelegenheid, 2008, 2019). The proposal put forward in 2008 presented a particularly detailed pension scheme that included actuarial rules for deferral of pension rights to a later retirement age. In this study we assume that this pension scheme is in operation and that, in hypothetical retirement scenarios, individuals who work beyond age 65 defer all or part of their state pension rights until retirement to smooth out their total income. The scheme allows an employee to defer all or part of his or her state pension rights for a maximum of five years. It is not possible to accrue additional rights during the deferral period. Deferred rights are actuarially increased at the time of claim. The increase is calculated as the product of the deferred pension rights at the statutory retirement age and an actuarial factor specific to the age at which the pension rights are ultimately claimed. The top panel of Table 1 shows the year-specific actuarial factors for possible ages of claiming state pension rights. The factor is calculated by dividing life expectancy at age 65 by life expectancy at age 65 less the number of years of delay. The life expectancy at age 65 is the average of the life expectancies of men and women at age 65. For example, the factor for claiming pension rights at age 70 is 1.354. It is calculated by applying the life expectancy at age 65 in 2007, which is equal to 19.14 years according to Statistics Netherlands, plus five years of delay due to deferral of the pension rights from age 65 until age 70.

An occupational pension scheme

In the Netherlands, employees are obliged to participate in an occupational old-age pension scheme. In most cases the scheme is of the defined benefit type. It is funded so that pensions are financed from the premiums paid by the participants in the past and from the returns on the invested premiums. It is an individual scheme, but for employees with a partner it incorporates a widow's pension and an orphan's pension for children up to an age threshold. We do not incorporate the widow's and orphan's

	Retirement	Actuarial factor						
	age	2007	2008	2009	2010	2011	2012	2013
State pension	66	1.055	1.055	1.054	1.054	1.035	1.053	1.053
scheme	67	1.117	1.116	Actuarial factor200820092010201120121.0551.0541.0541.0351.0531.1161.1141.1141.1121.1131.1841.1821.1811.1781.1791.2621.2581.2571,2531.2531.3501.3451.3431.3371.3390.7190.7190.7470.7490.7650.7650.7650.7890.7920.8150.8150.8350.8380.8710.8710.8850.8870.9320.9320.9400.9411.0001.0001.0001.0001.0751.0751.0661.0641.1591.1591.1591.1381.1341.2541.2541.2541.2181.2111.3601.3601.3601.3061.2961.4801.4801.4801.4031.390	1.113	1.112		
	68	1.186	1.184		1.177			
	69	1.264	1.262	1.258	1.257	1,253	1.253	1.251
	70	1.354	1.350	1.345	1.343	1.337	1.339	1.335
Occupational	60	0.719	0.719	0.719	0.719	0.747	0.749	0.752
pension scheme	70 1.354 1.3 60 0.719 0.7 61 0.765 0.7 62 0.815 0.8 63 0.871 0.8	0.765	0.765	0.765	0.789	0.792	0.794	
	62	0.815	0.815	0.815	0.815	0.835	2012 1.053 1.113 1.179 1.253 1.339 0.749 0.792 0.838 0.887 0.941 1.000 1.064 1.134 1.211 1.296 1.390	0.839
	63	0.871	0.871	0.871	0.871	0.885		0.889
	64	0.871 0 0.932 0	0.932	0.932	0.932	0.940	0.941	0.942
	65	1.000	1.000	1.000	1.000	1.000	1.000	1.000
	66	1.075	1.075	1.075	1.075	1.066	1.064	1.063
	67	1.159	1.159	1.159	1.159	1.138	1.134	1.132
	68	1.254	1.254	1.254	1.254	1.218	1.211	1.208
	69	1.360	1.360	1.360	1.360	1.306	1.296	1.292
	70	1.480	1.480	1.480	1.480	1.403	1.390	1.384

Table 1: Actuarial factors for earlier and later retirement than at age 65 for years from2007 to 2013

Notes: The actuarial factors of the occupational old-age pension scheme for 2007 are obtained from Stichting Pensioenfonds Zorg en Welzijn (2007). The figures of later years are obtained from the "Statuten en reglementen" of later years. The actuarial factors of the state pension scheme are the authors' calculations according to Ministerie van Sociale Zaken en Werkgelegenheid (2008) and year-specific average life expectancies for men and women available from Statistics Netherlands.

pensions in our calculations of pension benefits since they would involve another layer of complexity. Furthermore, in case of a divorce, the other spouse can be entitled to part of the accumulated pension, but we do not take this into account in our calculations. We base our analysis on the defined benefit scheme of Pensioenfonds Zorg en Welzijn (PFZW), the second largest pension fund in the Netherlands.

Employees pay premiums into their pension plan. These premiums are calculated as the product of the full-time equivalent, the contribution rate, and the premium base. The contribution rate is shared by the employee and the employer. The premium base is determined by the difference between the gross labor income and the state pension offset. Employees do not accrue pension over the state pension offset since they already pay premiums for the state pension. Table 2 shows the state pension offset amounts for the years 2007 to 2013.

The occupational pension is paid as an annuity after the statutory retirement age. It is calculated as the sum of the annual pension rights accrued over the years that employees accrue pension rights. In a given year, the accrued rights are calculated as

Year	Accrua	al rate (%)	State pension offset (€)		
	Born before 1950	Born in or after 1950	Born before 1950	Born in or after 1950	
2007	1.75	2.05	11,872	9,819	
2008	1.75	2.05	12,209	10,097	
2009	1.75	2.05	12,466	10,309	
2010	1.75	2.05	12,674	10,482	
2011	1.75	1.95	12,898	10,667	
2012	1.75	1.95	13,062	10,802	
2013	1.75	1.95	13,227	10,940	

Notes: Accrual rates may differ across pension funds. The figures are for PFZW. Accrual rates and state pension offset amounts are for the old-age pension scheme. State pension offset is also referred to as the "franchise". The accrual rate and state pension offset for 2007 are obtained from Stichting Pensioenfonds Zorg en Welzijn (2007). The figures of later years are obtained from the "Statuten en reglementen" of later years.

the product of the fulltime equivalent, the accrual rate, and the premium base. The accrual rate is the rate at which the pension rights build up. Table 2 shows the accrual rates for the years 2007 to 2013.

The scheme is an average salary scheme, with each year's salary contributing in the same way. Two specific issues regarding the calculation of the pension annuity need to be mentioned. First, the pension annuity depends on the domestic situation. If the participant is not single when claiming pension rights for the first time, the pension annuity applies and the participant's spouse is entitled to a survivor's pension when the participant dies. If the participant is single when claiming for the first time, the pension annuity is increased by a certain fraction. Second, pension funds aim to increase the pension annuities each year in line with wage inflation in the sectors in which they operate. The actual increase can be equal to, lower, or higher than the increase in the average wage, depending on the financial situation of the pension fund ('conditional indexation'). In our analysis we assume no increase in wages and no indexation. Our analysis can therefore be interpreted as an analysis of real wages under the assumption of full indexation and equality of wage and price inflation.

Claiming the pension annuity before or after the statutory retirement age has implications for the amount of the pension annuity. Deferring the claim of the pension annuity at the statutory retirement age affects the amount of the pension annuity when it is claimed in two respects. First, the annuity deferred at the statutory retirement age will increase due to the actuarial adjustment at the age the annuity is claimed. The increase is calculated as the product of the pension annuity at the statutory retirement age and an actuarial factor specific to the age at which the annuity is claimed. The actuarial factor depends on mortality rates and an interest rate. The actuarial factors applied in case of retirement after the statutory retirement age are higher than 1, leading to a higher deferred pension. Second, the pension annuity at the time it is claimed will increase due to the additional rights accrued from the statutory retirement age until the time the rights are claimed, and due to the actuarial adjustment of these rights at the time these rights are claimed.

Claiming the pension annuity earlier than the statutory retirement age affects the pension annuity as follows. First, since pension rights are only accumulated until the early retirement age, the pension annuity will be lower than that if pension rights were accumulated until the statutory retirement age. Second, the pension annuity at the early retirement age will fall due to the actuarial adjustment at this earlier time. The actuarial adjustment factors for early retirement are lower than 1 so that early retirement is effectively penalized. Table 1 shows the full set of year-specific actuarial factors for all retirement ages.

Note that the actuarial increase due to deferral of pension rights beyond age 65 is lower in the state pension scheme than in the occupational pension scheme since the actuarial factors in the state pension scheme are lower. That is because the actuarial factors of the state pension scheme are driven by mortality rates only, whereas those of the occupational pension scheme are driven by mortality rates as well as by returns on invested pension premiums. It seems difficult for the government to increase the actuarial factors to the level of those of the occupational pension fund because the state pension system is unfunded, meaning that the government cannot generate returns on the premiums.

The tax system

Every person who lives in the Netherlands and has some source of income is subject to income tax. The income tax is applied per individual, with some specific allowances that depend on household composition and income of other household members. Aside from this, every person is required to buy basic health insurance from a private insurer at a premium that is independent of income. An income-dependent subsidy or tax compensates low income individuals for the regulated income-independent premium of the basic health insurance cover; this is integrated with income tax and national insurance premiums. Details of the calculation of income after tax and health insurance premiums are provided in the Appendix.

3. Data

We use administrative data from PFZW, which offers occupational pension plans for people working in the health care sector. Data are available for clients who accrue or claim pension rights during the period of observation from January 2007 to December 2014. The clients are cohorts of individuals aged 55 years or older at any given year of the observation period. The data include information on age, gender, marital status, number of years of pension accrual, accrued pension rights, claimed pension rights, full-time equivalent, and wage income earned during the period of pension accrual.

The initial sample consists of 1,657,978 individuals, where, for each individual, multiple observations are available across the months of the years for which data are available. We apply a number of restrictions to the initial sample. First, in the data, individuals who retire before the statutory retirement age often make use of an early retirement scheme that offers generous pension provisions. In the Netherlands, early retirement schemes (VUT) have gradually been phased out in the past twenty years. Therefore, individuals who participate in an early retirement scheme are not considered in this study. This restriction leads to a sample of 1,540,774 individuals. Second, we require that individuals work full-time in their last job and subsequently retire full-time. This further restriction leads to a sample of 8,666 individuals. The large decrease in the number of individuals is due to the fact that many pension fund clients have not yet reached the retirement age, so for them retirement is not yet relevant. Third, we require that individuals work full-time during the main part of their career before their part-time or full-time retirement. Therefore, we restrict our analysis to individuals who have accrued pension rights on full-time pensionable salary for at least 35 years by the time they turn 65. We require fewer years of pension accrual for younger workers and for those who retire earlier than at age 65. This restriction leads to a sample of 3,313 individuals who constitute the baseline study sample.

We do not use individual data for the state pension income but assume that all individuals receive the universal flat-rate amount. We thus also assume that all individuals are insured for the maximum period of insurance and receive the full state pension amount. The state pension income amounts we consider are those effective in the observation year when individuals become entitled to the state pension.

Table 3 presents descriptive statistics for a number of background and labor market characteristics. The majority of the sample is above age 60 and is married or living with a partner. About 65% of the sample consists of men. About half of the sample earns a gross annual wage close to the average of \notin 47,422 in 2018 (OECD, 2019), and the other half earns a higher wage. Among the individuals who have accrued pension

Characteristic	Attribute	%
Age	55-60	21.92
	61-65	77.92
Gender	Male	64.70
Marital status	Married or living with partner	75.30
Annual gross income (in €)	< 35,000	8.05
	≥ 35,000-55,000	46.34
	≥ 55,000	45.16
Number of years of pension accrual	25-34	46.07
	35	53.93

Table 3: Background and labor market characteristics

Notes: The figures are based on 3,313 individuals. Annual gross income, expressed in euros, is the average wage earned in the last job in the sample. Totals may not add due to rounding.

rights for at least 25 years, about half have accrued pension rights for at least 35 years. The reason for the high fractions of older individuals, of men, and of high-income earners in the sample is that the sample is selected to consist of individuals with at least 25 years of pension accrual on full-time pensionable salary, whereas women often work part-time and earn lower wages and do not accrue (or accrue for fewer years) pension rights on full-time pensionable salary during their working career.

4. Method and assumptions

We analyze the financial well-being of older partially and fully retired workers in terms of their replacement rate, which we define as the ratio of retirement income over work income. Retirement income consists of the occupational pension income plus the state pension income if the individual has reached the statutory retirement age. In the case of partial retirement, retirement income consists of part-time work income plus part-time occupational pension income, but also of part-time state pension income if the individual has reached the statutory retirement age. Retirement income comes from pension savings and excludes any type of personal savings. In fact, in the Netherlands, many people make mortgage payments and therefore have less income available for consumption during their working years, whereas they finish their payments and thus have more income available for consumption during their consumption during their retirement years. This means that personal savings may affect the amount of income available for consumption both before and after retirement and hence also the replacement rate.

When calculating the replacement rate, we define work income as the wage earned in the last job before age 60. Alternatively, we could define it as the average of lifetime wages earned. The two income amounts will differ from each other to the extent that the age profile of wages slopes upward. However, work income affects replacement rate through the pension income in the numerator of the replacement rate and the work income in the denominator. This means that an alternative work income profile specification could have different effects on the replacement rates and it is not a priori clear which effect dominates. Throughout our analysis we use work income from the last job when calculating a replacement rate. A main economic motivation is that retired individuals may evaluate their retirement income adequacy by comparing their pension income against what they earned during the final years of their working career instead of the average of what they earned throughout their entire career. We approximate the wages earned during the final years of the career with the wage earned in the last job. This approximation is likely to be good for at least two reasons. First, age-income profiles in the Netherlands are fairly flat as from age 60 (Knoef et al., 2013). Second, job mobility is particularly low at the end of the working life, and therefore one's wage is unlikely to show large variation during the final years of the career (Euwals et al., 2014). Therefore, using the wage amount from the last job in the calculation of replacement rates may lead to replacement rate predictions that are accurate approximations of how individuals evaluate their retirement income adequacy. Besides this economic reason, it is difficult to consider

average lifetime work income in the calculation of a replacement rate because, in the data, complete histories of wages earned are available only for a limited number of individuals.

We calculate both gross and net replacement rates. The gross replacement rate uses the gross pension and work income amounts (amounts before tax). To calculate the net replacement rate, we apply the income tax rules to the gross pension and work income amounts, as described in Section 2. Even though gross (work) income remains the same, net income may change with age in the final years before full retirement, due to the change in marginal tax rates at the statutory retirement age, or due to the employed person's tax credit. By taking full-time work income before age 60 as the reference amount, we do not incorporate these changes in the replacement rates. The age-dependent "work bonus" may also change net work income in the final years before full retirement. This tax credit applied from 2009 until 2018, with an amendment in 2013. We do not account for this tax credit in our calculations because it does not apply to all cohorts of older workers in our data, and because it has been abolished and has no policy relevance.

We conduct three types of analyses. First, we analyze the actual replacement rates. We calculate the actual replacement rates using observed wages earned in the last job, occupational pension annuities received from the age individuals are retired, and the universal state pension annuity received if individuals have reached the state pension age. In the data very few individuals are observed to retire full-time after the statutory retirement age or to retire part-time from a fulltime job at any given age (Section 3). Therefore, we calculate the actual replacement rates only for the individuals who retired full-time at the statutory retirement age, and for a smaller group of individuals who retired full-time before the statutory retirement age.

Given that a small number of individuals retire full-time after the statutory retirement age, or retire part-time at any given age, we predict, in a second analysis, replacement rates in scenarios of partial and abrupt full retirement at all ages from 60 to 70. We predict the replacement rates using observed wages earned in the last job, occupational pension annuities predicted on the basis of the observed accrued pension rights of individuals in the data, and the universal state pension annuity received if individuals retire at the state pension age, and otherwise the state pension annuities predicted on the basis of the universal state pension annuity if individuals defer claiming state pension rights.

Occupational pension annuities in alternative retirement scenarios are predicted as follows. We start with the observed accrued pension rights of individuals who retired at the statutory retirement age, but also with those of the smaller group of individuals who retired before or after the statutory retirement age. Using the accrued pension rights observed, we predict pension annuities using the stylized pension rules described in Section 2 that apply to Stichting Pensioenfonds Zorg en Welzijn or to Stichting Pensioenfonds ABP, the two largest pension funds in the Netherlands. Note that, unlike earlier studies of retirement income adequacy, we do not make assumptions about wage rate, wage growth, and employment history of workers to forecast their pension annuities (cf., e.g., De Bresser and Knoef, 2015; Knoef et al., 2016). We observe accrued pension rights at the time of retirement, and use them to predict pension annuities in different retirement scenarios.

In an abrupt retirement scenario where a hypothetical individual works full-time until age 70, we predict the pension annuity at age 70 as follows. First, the pension rights accrued until age 65 and deferred at this age until age 70 increase due to the actuarial adjustment at age 70, when these rights are ultimately claimed. We calculate this actuarial adjustment by multiplying the accrued pension rights at age 65 with the actuarial factor specific to age 70, which is, for example, 1.480 in 2010 (Table 1). Second, due to deferring the claim of pension rights and working during the years from age 65 until age 70, additional pension rights are accrued during these years; these are, again, actuarially adjusted at age 70. Accrued pension rights are calculated by adding up the rights accrued in each year. In a given year, accrued rights are calculated as the product of the full-time equivalent, the accrual rate, and the premium base. The full-time equivalent is equal to 1 if the individual is assumed to be working full-time. The accrual rate is, for example, 2.05 in 2010 (Table 2). The premium base is determined by the difference between gross income and the state pension offset. We consider the observed gross wage income earned in the last job as the gross income. The state pension offset is equal to €10,482 in 2010. We then multiply the pension rights accrued from age 65 to 70 with the actuarial factor specific to age 70, which is 1.480. The sum of the pension rights deferred at age 65 and actuarially adjusted at age 70 and the additional pension rights accrued during the deferral years from age 65 until 70 and actuarially adjusted at age 70 determines the predicted pension annuity at age 70.

In a partial retirement scenario, where the hypothetical employee works full-time until, for example, age 65, and partially retires from age 65 to 70, we predict the pension annuity at age 70 in the same manner as in the abrupt retirement scenario, except that the full-time equivalent during the accrual of pension rights from age 65 to 70 is considered as 0.5.

The state pension annuity is given by the universal flat-rate state pension paid to all individuals at the state pension age. State pension annuity is predicted if state

pension rights are deferred at the state pension age. This prediction is done in the same manner as for the occupational pension annuity except that, according to the assumed state pension scheme that allows an employee to defer part or all of his or her state pension rights, suitable actuarial factors are used, and it is not possible to accrue additional pension rights during the deferral period (Section 2). For example, in an abrupt retirement scenario where a hypothetical individual works full-time until age 70, we predict the pension annuity at age 70 as follows. The pension rights accrued until age 65 and deferred at this age until age 70 increase due to the actuarial adjustment at age 70, when these rights are ultimately claimed. We calculate this actuarial adjustment by multiplying the accrued pension rights at age 65 with the actuarial factor specific to age 70, which is, for example, 1.343 in 2010 (Table 1).

Given the predicted replacement rates based on the actual earnings and accrued pension rights of individuals in the sample data, we study, in a third analysis, the distribution of the replacement rates and how partial retirement, compared to full retirement, affects the risk of falling below a minimum target replacement rate. In the Netherlands, social insurance schemes, including the unemployment and disability insurance schemes, pay benefits that replace 70% of the gross wage earned in the last job. We assume that the statutory replacement rate in social insurance schemes is also the target replacement rate in the pension domain, so that pension income from the state and occupational pension schemes replace at least 70% of the employee's final gross earnings. However, the net replacement rate is higher than the gross replacement rate. We abstract from the impact of income taxation and maintain that the minimum target replacement rate is also 70% in net terms. We present results based on a minimum target replacement rate of 80% as part of a sensitivity analysis.

5. Results

Observed replacement rates

Figure 1 presents the average of the gross replacement rates at age 65 when individuals retire full-time at age 60 to 65. The figure shows that individuals who retire at age 63 or 64 accumulate the most pension rights relative to their final full-time wage compared to other individuals, who retire at an earlier or later age. A possible reason for this is that those who retire at age 63 or 64 accumulate more pension rights to afford retirement at such age. This result suggests that individuals aim at achieving a level of retirement income that they consider adequate. As a result, it also suggests that when deciding on the retirement age, retirement income adequacy plays a greater role than institutional incentives or social norms, which call for retirement at the statutory retirement age. This finding might explain why the effective retirement age is lower than the statutory retirement age in the Netherlands, same as in many other Western countries.



Figure 1: Observed mean gross replacement rate at age 65 when individuals retire fully from a fulltime job at given ages from 60 to 65. Replacement rates are based on wages earned in the last job and on pension rights accrued on full-time pensionable salary for at least 35 years by the time of retirement at age 65. At least 30 years of pension accrual are required for workers who retire at age 60. The mean replacement rates at given ages from 60 to 65 are calculated based on data for 2,711 individuals.

Predicted replacement rates in scenarios of full and partial retirement

Figure 2 presents the gross and net replacement rate trajectories in scenarios of abrupt retirement at age 60 to 70. For example, in the left panel of the figure, in the scenario of abrupt retirement at age 60, the gross replacement rate is 33% from age 60 until 65, and 54% from age 65 and onwards. A replacement rate at a given age represents the average of the replacement rates predicted at that age, based on the actual wages earned in the last job, and the pension rights accrued by individuals who retired fully from a full-time job at age 54 to 65. Pension rights are accrued on full-time pension-able salary for at least 35 years by the time these individuals who retire at age 60.

A comparison of gross and net replacement rates shows that net replacement rates are higher than gross replacement rates. This is because pension income is lower than wage income, meaning that pension income is taxed in a lower tax bracket. The difference between gross and net replacement rates is higher after age 65 than before age 65. This is because the tax rate at age 65 is lower than that before age 65 (Section 2).

The gross replacement rate for abrupt retirement at age 65 is 77%. This is substantially below the gross replacement rate of 97% calculated by the OECD using the national parameters and rules applying in 2016 (OECD, 2017). The difference is mainly due to the fact that the OECD calculates the replacement rate based on a fictitious person who earns the median income during his entire career and accrues pension rights for 45 years on a full-time pensionable salary. In the sample data, however, very few people accrue pension rights for a total of 45 years on full-time pensionable salary. For example, among those observed who had worked full-time and retired the year after, the mean number of years of pension accrual on full-time pensionable salary is 26.6 years for men and 21.6 years for women. The replacement rates presented are calculated, however, based on the restriction of at least 35 years of pension accrual on fulltime pensionable salary. Indeed, Knoef et al. (2016) argue that, when evaluating the effectiveness of the Dutch pension system to replace pre-retirement earnings, attention should be given to the fact that career interruptions limit the amount of pension rights that individuals accrue. In the next section we analyze how different numbers of years of pension accrual affect retirement income adequacy when individuals retire part-time and full-time.

As to the gross replacement rates attained at the time of retirement, we see a large difference between when people retire before the statutory retirement age, and when they retire at or after the statutory retirement age. While the gross replacement rates range from 33% to 52% when people retire before the statutory retirement age, they

are much higher, ranging from 77% to 98%, when they retire at or after the statutory retirement age. This is due to the fact that, before reaching the statutory retirement age, individuals do not have access to the state pension income, while occupational pension income replaces their final wage only to a limited extent, especially because occupational pension rights are actuarially penalized when claimed earlier than the statutory retirement age.

As to the gross replacement rates attained at the age of 70, we see large differences across people who retire at different ages from 60 to 70. That is, at age 70, the gross replacement rates range from 54% to 72% across people who retire at the age of 60 to 64, while they range from 77% to 98% across people who retire at the age of 65 to 70. This shows that delaying retirement increases retirement income adequacy, and the increase after age 65 is higher than that before age 65. This is due to the fact that the actuarial increase in pension rights when delaying retirement is higher at older ages. Another finding is that pension income does not fully replace the final earnings before retirement even when retirement is delayed until age 70. For example, for the individual who retires at the standard retirement age of 65, pension income replaces about 77% of the final earnings, and it takes five additional years of full-time work to replace about 98% of the final earnings. A contributing factor is that gross replacement rates are capped at 100% by law, leading to mean replacement rates below 100%.

The differences across the net replacement rates are smaller compared to the differences across the gross replacement rates, especially after age 65. This is due to the progressive nature of the tax system. A person who retires later and accrues more pension rights is taxed at higher rates and will have a lower replacement rate. This demonstrates the redistributive role of the progressive tax system in pension provision.

Figure 3 presents the gross and net replacement rate trajectories in scenarios of partial retirement at given ages from 60 to 65. The duration of partial retirement is always considered to be five years.

Figure 4 reproduces the net replacement rates from Figures 2 and 3. In particular, the left panel shows the net replacement rates where individuals retire full-time from their full-time job between age 60 to 65, and the right panel shows the net replacement rates where individuals retire part-time from their full-time job at the same ages for a period of five years. A comparison of the replacement rates in the left and right panels of the figure leads to several findings.

First, considering the replacement rates attained before age 65, when individuals retire part-time, they attain replacement rates that are about 30 percentage points



Figure 2: Predicted mean gross (left panel) and net (right panel) replacement rates in retirement scenarios where individuals retire fully at given ages from 60 to 70. Replacement rates are based on actual wages earned in the last job and on pension rights accrued on full-time pensionable salary for at least 35 years by the time of retirement at age 65. At least 30 years of pension accrual are required for workers who retire at age 60. A mean replacement rate at a given age is predicted using data for 3,313 individuals who retired fully from a full-time job at given ages from 54 to 65.



Figure 3: Predicted mean gross (left panel) and net (right panel) replacement rates in retirement scenarios where individuals retire part-time from their full-time job for a period of five years at given ages from 60 to 65. During partial retirement, individuals work 50% of the hours they worked in their full-time job and claim 50% of their accrued occupational and state pension rights. Replacement rates are based on actual wages earned in the last job and on pension rights accrued on full-time pensionable salary for at least 35 years by the time of retirement at age 65. At least 30 years of pension accrual are required for workers who retire at age 60. A mean replacement rate at a given age is predicted using data for 3,313 individuals who retired fully from a full-time job at given ages from 54 to 65.

higher than when they retire full-time at the same ages. Considering the level of replacement rates, partial retirement allows individuals to attain at least the minimum target replacement rate of 70%, while full retirement at the same ages leads to replacement rates that are below 56%. This shows to what extent partial retirement helps workers to ensure their financial well-being against the large income shock they face if they retire before the statutory retirement age and have to rely on their occupational pension income.

Second, considering the replacement rates attained after age 65, retiring parttime instead of full-time before age 65 leads to replacement rates that are about 15



Figure 4: Income smoothing with partial retirement. Predicted mean net replacement rates in retirement scenarios where individuals retire from their full-time job fully (left panel reproducing the right panel of Figure 2 for select retirement scenarios) and partially (right panel reproducing the right panel of Figure 3) at given ages from 60 to 65.

percentage points higher. Compared to the much larger gap of 30 percentage points before age 65 presented above, the smaller gap after age 65 is due to the effect of the first pillar pension in the net replacement rate.

Overall, compared to full retirement, partial retirement results in a much smoother income path. For example, for someone who retires fully at age 63, the average net replacement rate moves from 50% at age 63 to 83% at age 70, while for someone who partially retires at age 63 for a period of five years, the average net replacement rate moves from 78% at age 63 to 96% at age 70. This demonstrates to what extent partial retirement helps individuals to smooth their income path over the course of their retirement when they stop working before the normal retirement age.

The risk of falling below the minimum target replacement rate

In Figures 2 and 3 we analyze replacement rates in scenarios of partial retirement and abrupt full retirement at ages from 60 to 70. In these scenarios, a given replacement rate represents the mean of the replacement rates predicted on the basis of the actual earnings and accrued pension rights of the individuals in the sample. Here we study the distribution of the predicted replacement rates, and in particular we analyze the fractions of individuals who fall below the minimum target replacement rate of 70% when they retire part-time compared to full-time. In this analysis we focus on retirement before the statutory retirement age since after that age the state pension largely enables workers to attain the minimum target replacement rate, as shown in the preceding subsection.

In Figure 5 we analyze the fractions of individuals with a net replacement rate below the minimum target replacement rate at the age (from 60 to 65) when they retire full-time and part-time (any number of years). The figure shows that almost all



Figure 5: Fractions of individuals with a net replacement rate below the minimum target replacement rate at the age (from 60 to 65) when they retire full-time and part-time (any number of years). Replacement rates are based on actual wages earned in the last job and on pension rights accrued on full-time pensionable salary for at least 35 years by the time of retirement at age 65. At least 30 years of pension accrual are required for workers who retire at age 60. Replacement rates at a given age are predicted using data for 3,313 individuals who retired fully from a full-time job at given ages from 54 to 65.

individuals who retire full-time before the age of 65 fall below the minimum target replacement rate. As demonstrated by Figure 2, this is due to the fact that, before the statutory retirement age, occupational pension income replaces the final wage to a limited extent, especially because occupational pension rights are actuarially penalized when claimed earlier than the statutory retirement age. The high risk of falling below the minimum target replacement rate before the statutory retirement age can explain why individuals often retire at the statutory retirement age, when they are entitled to the state pension benefit (Mastrobuoni, 2009; Atav et al., 2019). The figure shows to what extent partial retirement helps workers to insure their financial well-being against this high risk. That is, based on the replacement rates predicted among the individuals in the sample, partial instead of full retirement substantially decreases the fraction of individuals who fall below the minimum target replacement rate, for example from about 95% to only 12% at age 63.

In Figure 6 we analyze the fractions of people with a net replacement rate below the minimum target replacement rate at age 70 when they retire full-time and part-time at each age from 60 to 65. We distinguish among different partial retirement scenarios, where individuals spend different numbers of years in partial



Figure 6: Fraction of individuals with a net replacement rate below the minimum target replacement rate at age 70 when individuals retire full-time and part-time – with different numbers of years spent in part-time work – at given ages from 60 to 65. Replacement rates are based on actual wages earned in the last job and on pension rights accrued on full-time pensionable salary for at least 35 years by the time of retirement at age 65. At least 30 years of pension accrual are required for workers who retire at age 60. Replacement rates at a given age are predicted using data for 3,313 individuals who retired fully from a full-time job at given ages from 54 to 65.

retirement. We analyze the risk of falling below the minimum target replacement rate after the statutory retirement age, in particular at age 70. Thus, in all scenarios the state pension income is taken into account in the replacement rates. There are two main findings. First, the risk of falling below the minimum target replacement rate decreases as people retire full-time or part-time at later ages. This is because accrued pension rights are higher at later retirement ages, and because pension rights are less heavily penalized at later retirement ages. Second, partial rather than full retirement substantially decreases the risk of falling below the minimum target replacement rate. For example, at age 63, while this risk is about 22% when individuals retire full-time, it is about 14% when they instead remain employed part-time for two years and about 4% when they remain employed part-time for five years.

Vulnerable groups

In Figure 7 we analyze the risk of falling below the minimum target replacement rate at age 70 among the individuals who accumulated pension rights for at least 35 years (the same as those in Figure 6) and among those who accumulated pension rights for 25 to 34 years. Among the individuals who accumulated pension rights for fewer years, the risk of falling below the minimum target replacement rate is substantially higher at any given age at which they retire full-time or part-time. However, the risk is substantially lower among the individuals who retire part-time than among those who retire full-time, at any given age. In fact, individuals who retire part-time for a period of five years and have 25 to 34 years of pension accrual appear to have a very similar risk profile as those who retire full-time and have at least 35 years of pension accrual.

In Figure 8 we analyze the impact of the earnings level on the risk of falling below the minimum target replacement rate. We consider two income groups with respect to the amount of wages earned in the last full-time job: €25,000-34,999 and €55,000-64,999. High-income earners appear to be much more likely to have a replacement rate that is below the minimum target level. The main reason is that, for a high-in-come earner, the state pension income is much lower relative to the wage income.

Figure 9 shows how the domestic situation affects the risk of falling below the minimum target replacement rate. Those living with a partner are much more likely to fall below the minimum target replacement rate. There are two main reasons for this. First, the state pension is lower for those living with a partner. Second, the gross wage earned in the last full-time job is lower for singles than for those living with a partner. In the sample, while singles earned $\leq 48,440$ on average, those living with a partner earned $\leq 64,446$ in the last job where they worked on a full-time basis.

Figure 10 shows how the risk of falling below the minimum target replacement rate differs between men and women. Compared to women, men are roughly twice more likely to fall below the minimum target replacement rate regardless of the age at which they retire full-time or part-time. Higher replacement rates for women reflect the fact that women earn less and that for them the state pension replaces a much higher share of their pre-retirement earnings. This result is remarkable because, despite the fact that we require that women, same as men, accrue occupational pension rights on a full-time pensionable salary for at least 30 years by the time they retire at age 60, or for more years if they retire later, their wage is low enough that it leads to replacement rates roughly twice as high as those of men. In the sample data, while women earned ξ_{48} ,860 on average in the last job where they worked on a full-time basis, men earned $\xi_{65,551}$. The wage differential between women and men is due to the fact that, in the health care sector, men more often occupy higher paid medical professions.



Figure 7: Fraction of individuals with a net replacement rate below the minimum target replacement rate at age 70 when individuals retire full-time and part-time – with different numbers of years spent in part-time work – at given ages from 60 to 65: for individuals who accrue pension rights for at least 35 years (left panel reproducing Figure 6) and for those who accrue pension rights for 25 to 34 years (right panel) by age 65. At least 30 years of pension accrual are required for workers who retire at age 60. Replacement rates are based on actual wages earned in the last job and on pension rights accrued on full-time pensionable salary for the studied cases of number of years of pension accrual. For individuals with pension accrual for at least 35 years, replacement rates at a given age are predicted using data for 3,313 individuals who retired fully from a full-time job at given ages from 54 to 65. For individuals with pension accrual for 25 to 34 years, replacement rates at a given age are predicted using data for 1,716 individuals who retired fully from a full-time job at ages from 54 to 64.



Figure 8: Fractions of individuals with a net replacement rate below the minimum target replacement rate at age 70 when individuals retire full-time and part-time – with different numbers of years spent in part-time work – at given ages from 60 to 65: for individuals with earnings of \pounds 25,000– \pounds 34,999 (left panel) and for those with earnings of \pounds 55,000– \pounds 64,999 (right panel). Earnings represent wage income in the last full-time job before (partial) retirement. Replacement rates are based on actual wages earned in the last job and on pension rights accrued on full-time pensionable salary for at least 35 years by the time individuals retire at age 65. At least 30 years of pension accrual are required for workers who retire at age 60. For low income earners, replacement rates at a given age are predicted using data for 323 individuals who retired fully from a full-time job at ages from 54 to 64. For high-income earners, replacement rates at a given age are predicted using data for 539 individuals who retired fully from a full-time job at ages from 54 to 64.



Figure 9: Fraction of individuals with a net replacement rate below the minimum target replacement rate at age 70 when individuals retire full-time and part-time – with different numbers of years spent in part-time work – at given ages from 60 to 65: for individuals with a partner (left panel) and for singles (right panel). Replacement rates are based on actual wages earned in the last job and on pension rights accrued on full-time pensionable salary for at least 35 years by the time individuals retire at age 65. At least 30 years of pension accrual are required for workers who retire at age 60. For individuals with a partner, replacement rates at a given age are predicted using data for 2,422 individuals who retired fully from a full-time job at ages from 54 to 64. For singles, replacement rates at a given age are predicted using data for 891 individuals who retired fully from a full-time job at ages from 54 to 65.



Figure 10: Fractions of individuals with a net replacement rate below the minimum target replacement rate at age 70 when individuals retire full-time and part-time – with different numbers of years spent in part-time work – at given ages from 60 to 65: for men (left panel) and women (right panel). Replacement rates are based on actual wages earned in the last job and on pension rights accrued on full-time pensionable salary for at least 35 years by the time individuals retire at age 65. At least 30 years of pension accrual are required for workers who retire at age 60. For men, replacement rates are predicted using data for 2,293 individuals who retired fully from a full-time job at ages from 54 to 65. For women, replacement rates are predicted using data for 1,020 individuals who retired fully from a full-time job at ages from 54 to 64.

Policy simulations

Using the actual earnings and accrued pension rights of the individuals in the sample, we predicted replacement rates in alternative retirement scenarios. Using the distribution of the predicted replacement rates, we studied the fraction of individuals who fall below a minimum target replacement rate of 70% at age 70. In Figure 6 we showed how this fraction depends on an early retirement age and the type of

retirement (partial or full). Here we consider a higher minimum target replacement rate, and study the fraction of individuals who fall below the higher minimum target replacement rate, compared to the fraction of individuals who fall below the minimum target replacement rate of 70%. Figure 11 presents the risk of falling below the minimum target replacement rate when the minimum is defined as 70% and when it is defined as 80%. In the case of full retirement, the risk of falling below the minimum target replacement rate at age 70 increases substantially, by some 25 percentage points, when the threshold for the minimum target replacement rate is increased from 70% to 80%. In the case of partial retirement at given ages from 60 to 62, we obtain a similar increase. The increase is smaller at later ages of partial retirement. Another notable finding is that individuals who retire part-time for a period of five years and fall below the minimum target replacement rate of 80% have a similar risk profile as individuals who retire full-time and fall below the minimum target replacement rate of 70%. This shows that individuals who retire part-time instead of full-time before the statutory retirement age more often attain a retirement income that is deemed adequate from the point of view of the statutory gross replacement rate of 70% provided in social insurance schemes.

Throughout our analyses we assumed zero wage inflation, and therefore we did not consider indexation of occupational pension rights when making projections in hypothetical retirement scenarios. We also assumed zero price inflation. Here we retain our assumption of no wage inflation, and hence no indexation, since the largest pension funds did not index pension rights during the past five to ten years due to their low funding ratios. However, we allow for price inflation at an annual rate of 1%. That is, we deflate the pension rights paid during each year of retirement to obtain corresponding amounts in real terms. Figure 12 shows the risk of falling below the minimum target replacement rate at age 70 when paid pension rights are not adjusted, and when they are adjusted for price inflation. The figure shows that the risk of falling below the minimum target replacement rate is higher due to price inflation since deflated pension rights are smaller than nominal pension rights, relative to final earnings. For example, in the case of full retirement at age 60, the risk is about 10 percentage points higher when pension rights are adjusted for price inflation compared to when they are not adjusted. The large difference is due to the fact that individuals who retire full-time at age 60 are exposed to 1% price inflation and no indexation for ten years until retirement at age 70. The risk becomes smaller as individuals delay retirement since they accumulate more pension rights and defer their pension claims until retirement. The risk is substantially smaller in the case of partial

retirement. This shows that individuals who retire part-time instead of full-time are much less prone to changes in the general price level over the retirement period.

In 2014 the "Generation Pact" was introduced in the collective labor agreements of a number of municipalities. In subsequent years the scheme was expanded to other sectors. The Generation Pact is a partial retirement scheme that offers two types of financial incentives that make it more attractive than the partial retirement scheme already offered by most pension funds. First, it allows the employee to accrue pension rights based on the former wage instead of the part-time wage earned during partial retirement. Second, the wage rate during partial retirement is higher than that in the former job. It is assumed that the employer will finance the higher wage of the older worker by employing a younger worker, who will replace the hours released by the older worker but earn an hourly wage that is lower than that earned by the older worker. For example, in a widely used variant of the scheme, an employee works 80% of the hours he worked in his former full-time job, earns 90% of the wage he earned in his former full-time job, and accrues pension rights based on 100% of the wage he earned in his former full-time job. Alternative variants are possible, such as that the number of work hours is 60% of the hours worked in the former full-time job, the wage earned is 80% of the wage earned in the former full-time job, and pension rights are accrued based on 100% of the wage earned in the former full-time job. However, the employee must work at least 50% of his former number of work hours and start to participate in the scheme nor more than ten years before the statutory retirement age.

Here we analyze the replacement rates of workers when they participate in the Generation Pact scheme and earn wages and build pension rights on favorable terms, and when they participate in the default partial retirement scheme offered by their pension fund. Our particular aim is to show how much the final pension income is affected if employees participate in the Generation Pact and rely on a favorable wage income from their employer, compared to if they participate in the default partial retirement scheme and rely on their wage income as well as on their occupational pension income.

During participation in either scheme, we assume that individuals work 80% of the hours they worked in their former full-time jobs. We consider the widely used variant of the Generation Pact, where individuals are paid 90% of the wage they earned in their former full-time job, and accrue pension rights based on 100% of the wage they earned in their former full-time job. In the default partial retirement scheme, however, individuals are paid 80% of their former wage, claim occupational pension rights equal to 10% of their former wage, and accrue pension rights based on the part-time wage they earn during partial retirement. This means that during participation in either scheme, the total income received is equal to 90% of the former wage earned, but this income is paid by the employer when participating in the Generation Pact scheme, while it is financed by labor income as well as by accrued pension rights when participating in the default partial retirement scheme.

Figure 13 presents the net replacement rates of the individuals when they participate in the Generation Pact and when they participate in the default partial retirement scheme. Considered are scenarios where individuals retire part-time from their full-time job at given ages from 60 to 64 until the statutory retirement age. During partial retirement, individuals attain a net replacement rate of about 96% when they participate in either of the two schemes. Considering the scenario of partial retirement at age 60, individuals attain, upon full retirement from age 65, a net replacement rate of about 85% if they participate in the default partial retirement scheme, while they attain a net replacement rate of about 92% if they participate in the Generation Pact. The difference is due to the fact that individuals accrue pension rights on a lower wage income when they participate in the default partial retirement scheme. The difference reduces to 1 percentage point when individuals partially retire at age 64. The reduction is caused by the fact that individuals accrue more pension rights if they partially retire at a later age. This shows that the gain in income replacement in retirement from participating in the Generation Pact is higher when employees partially retire earlier.

Figure 14 shows the risk of falling below the minimum target replacement rate of 70% at age 70 when individuals participate in the default partial retirement scheme and when they participate in the Generation Pact. Considering the scenario of partial retirement at age 60, the risk is about 19% if individuals participate in the default partial retirement scheme, while it is about 8% if they participate in the Generation Pact. The risk becomes smaller if individuals partially retire later in the default partial retirement scheme.

Section 2 described an alternative state pension scheme proposed by the Dutch government in 2008. The scheme allows full or partial deferral of state pension rights to a later retirement age. In our analyses, we assumed that the proposed pension scheme is in operation, and in hypothetical retirement scenarios individuals who work beyond age 65 defer all or part of their state pension rights until retirement to smooth their total income. Here we assume that the effective state pension scheme is in operation and that deferral of state pension receipt is thus not possible. However, we consider that individuals who delay retirement defer their occupational pension rights as the occupational pension scheme allows. Considering the net replacement rates attained at age 70, we find small differences between the situation where individuals delay receipt of their state pension rights and where they claim their state pension rights at the statutory retirement age. For example, when an individual partially retires at age 63 and defers the state and occupational pension rights for five years until age 68, such person attains a net replacement rate of about 96% at age 70, whereas when partially retiring at age 63 for five years and deferring the occupational pension rights until age 68 but claiming state pension rights at the statutory retirement age, such person attains a net replacement rate of about 95% at age 70. As to the risk of falling below the minimum target replacement rate at age 70, this risk is 4.8% for the former group and 4.1% for the latter group.



Figure 11: Fractions of individuals with a net replacement rate below 70% (left panel reproducing Figure 6) and 80% (right panel) at age 70 when individuals retire full-time and part-time – with different numbers of years spent in part-time work – at given ages from 60 to 65. Replacement rates are based on actual wages earned in the last job and on pension rights accrued on full-time pensionable salary for at least 35 years by the time individuals retire at age 65. At least 30 years of pension accrual are required for workers who retire at age 60.



Figure 12: Fractions of individuals with a net replacement rate below the minimum target replacement rate at age 70 when individuals retire full-time and part-time – with different numbers of years spent in part-time work – at given ages from 60 to 65: when paid pension rights are not adjusted (left panel reproducing Figure 6) and when they are adjusted (right panel) for 1% annual price inflation. Replacement rates are based on actual wages earned in the last job and on pension rights accrued on full-time pensionable salary for at least 35 years by the time individuals retire at age 65. At least 30 years of pension accrual are required for workers who retire at age 60.



Figure 13: Predicted mean net replacement rates when individuals participate in the default partial retirement scheme (left panel) and when they participate in the Generation Pact scheme (right panel). Presented are partial retirement scenarios where individuals retire part-time from their full-time job at given ages from 60 to 64 until the statutory retirement age. When participating in the default partial retirement scheme, individuals work 80% of the hours they worked in their full-time job, earn 80% of their former full-time wage, claim occupational pension rights that amount to 10% of their former full-time wage, and accrue occupational pension rights based on their part-time wage. When participating in the Generation Pact scheme, individuals work 80% of their former full-time job, earn 90% of their former full-time wage, do not claim occupational pension rights, and accrue occupational pension rights based on their former full-time pensional pension rights based on their former full-time pensionable salary for at least 35 years by the time of retirement at age 65. At least 30 years of pension accrual are required for workers who retire at age 60.



Figure 14: Fractions of individuals with a net replacement rate below 70% when they participate in the default partial retirement scheme (left panel) and when they participate in the Generation Pact scheme (right panel) at age 70, when they retire part-time at given ages from 60 to 64 until the statutory retirement age. Replacement rates are based on actual wages earned in the last job and on pension rights accrued on full-time pensionable salary for at least 35 years by the time of retirement at age 65. At least 30 years of pension accrual are required for workers who retire at age 60.

6. Conclusion

Policymakers are confronted with the challenge of ensuring an adequate retirement income for older persons, while keeping the pension system sustainable. Partial retirement can be an instrument to achieve both goals. Combining wage and pension income can fund old-age consumption, and it can induce individuals to extend their working lives up to the statutory retirement age or beyond. We have shown that, on average, older workers attain at least a minimum target replacement rate of 70% during partial retirement, and a higher replacement rate during full retirement, when they retire part-time and combine part-time earnings with part-time pension before the statutory retirement age. The replacement rates fall considerably below the minimum target replacement rate when workers retire full-time before the statutory retirement age. Furthermore, the distribution of replacement rates across individuals shows that those who retire part-time have a much smaller risk of falling below the minimum target replacement rate compared to those who retire full-time, both before and after the statutory retirement age. An important implication of this result is that, when individuals retire part-time instead of full-time at a given age, they are always less prone to shocks in pension entitlements due to, for example, zero indexation of occupational pension rights or a lower occupational pension income due to having accumulated pension rights for fewer years.

The relevance of partial retirement for pension income adequacy before the statutory retirement age is likely to increase in the coming years. Early retirement schemes have been phased out in the past ten years. This may increase the scope of partial retirement before the statutory retirement age as older workers now lack the possibility of early retirement from generous early retirement schemes (Bloemen et al., 2016; Muns, 2018). Furthermore, the statutory retirement age is being increased since 2013. Partial retirement can be expected to become prevalent among older workers, at least among those working in physically demanding occupations, who would therefore wish to retire earlier than the statutory retirement age (Vermeer et al., 2016; Kok et al., 2018). In fact, since 2014, a growing number of collective labor agreements allow older workers to participate in the "Generation Pact" scheme and retire partially, where they earn wages and build pension rights on favorable terms. From a public finances perspective, partial retirement can in fact be beneficial, as it increases the number of years of pension contributions and tax revenues during the years of otherwise early retirement. Indeed, it has been shown that partial retirement can increase total labor supply if it is offered to individuals who would otherwise fully retire in an abrupt manner before the statutory retirement age (Been et al., 2018).

We analyzed replacement rates in alternative retirement scenarios, based on the pension rights accrued and wages earned by individuals who work in the healthcare sector. Limiting our analysis to this specific labor market sector does not necessarily undermine the representative character of our analysis. Every retiree who never lived abroad receives the full state pension, and the majority of them participate in a mandatory occupational pension scheme of the defined benefit type. Furthermore, the occupational pension regulation described is very similar across pension funds. This means that our predictions of replacement rates would not have been different by any substantial amount if additional data, or data from other pension funds, were used. In fact, the mean gross and net replacement rates that we predict are comparable to those predicted by Knoef et al. (2016). Specifically, while we predict that the mean gross and net replacement rates are 77% and 87% (Figure 2), respectively, Knoef et al. predict that they are 71% and 84% (Tables 4 and 6). The analysis by Knoef et al. is based on occupational pension entitlements projected by Statistics Netherlands, using individual data from pension funds that are required to provide Statistics Netherlands with pension entitlement data on an annual basis.

References

- Alessie, R., Kapteyn, A., 2001. Savings and pensions in the Netherlands. Research in Economics 55 (1), 61–82.
- Atav, T., Jongen, E., Rabaté, S., 2019. The effects of the increase in the retirement age in the Netherlands. CPB Discussion Paper.
- Been, J., Kantarcı, T., van Vuuren, D., 2018. Increasing the labor force participation among older workers in the Netherlands: effects of the pension incentives, increasing retirement age, and partial retirement. Netspar Discussion Paper 05/2018–047.
- Belastingdienst, 2019. Loonheffingen. Handboek.
- Bloemen, H., Hochguertel, S., Zweerink, J. R., 2016. Gradual retirement in the Netherlands: an analysis using administrative data. Research on Aging 38 (2), 202–233.
- Cahill, K. E., Giandrea, M. D., Quinn, J. F., 2006. Retirement patterns from career employment. The Gerontologist 46 (4), 514–523.
- Chen, Y.–P., 1996. The role of the fourth pillar in the redesign of social security. The Geneva Papers on Risk and Insurance 21 (81), 469–477.
- De Bresser, J., Knoef, M., 2015. Can the Dutch meet their own retirement expenditure goals? Labour Economics 34, 100–117.
- European Commission, 2012. White paper. An agenda for adequate, safe and sustainable pensions. COM (2012) 55 final, February 16, 2012, Brussels.
- Euwals, R., Boeters, S., Bosch, N., Deelen, A., ter Weel, B., 2014. Employability and the labour market for older workers in the Netherlands. Netspar Panel Paper 43.
- Geppaart, C., 2010. Belastingwetten. Kluwer-Deventer.
- Kantarcı, T., Smeets, I. A. J., van Soest, A., 2013. Implications of full and partial retirement for replacement rates in a defined benefit system. Geneva Papers on Risk and Insurance Issues and Practice 38 (4), 824–856.
- Knoef, M., Alessie, R., Kalwij, A., 2013. Changes in the income distribution of the Dutch elderly between 1989 and 2020: a dynamic microsimulation. Review of Income and Wealth 59 (3), 460–485.
- Knoef, M., Been, J., Alessie, R., Caminada, K., Goudswaard, K., Kalwij, A., 2016. Measuring retirement savings adequacy: developing a multi-pillar approach in the Netherlands. Journal of Pension Economics and Finance 15 (1), 55–89.
- Kok, L., Lammers, M., van Soest, A., ter Weel, B., 2018. Vroegpensioenregelingen voor zware beroepen. Economische Statistische Berichten 103 (4758).
- Laczko, F., 1988. Partial retirement: An alternative to early retirement? A comparison of phased retirement schemes in the United Kingdom, France and Scandinavia. International Social Security Review 41 (2), 146–169.
- Mastrobuoni, G., 2009. Labor supply effects of the recent social security benefit cuts: Empirical estimates using cohort discontinuities. Journal of Public Economics 93 (11–12), 1224–1233.
- Ministerie van Sociale Zaken en Werkgelegenheid, 2008. Beslispunten flexibilisering AOW. Nota. Ministerie van Sociale Zaken en Werkgelegenheid, 2019. Principeakkoord vernieuwing
 - pensioenstelsel.
- Muns, S., 2018. Deeltijdpensioen kan fiscaal ongunstig zijn. Economische Statistische Berichten 103 (4767).
- OECD, 2017. Pensions at a Glance 2017: OECD and G20 Indicators. OECD Publishing, Paris.
- OECD, 2019. OECD Economic Outlook, Volume 2019 Issue 1. OECD Publishing, Paris.

- Poterba, J. M., 2014. Retirement security in an aging population. American Economic Review 104 (5), 1–30.
- Robinson, C., Clark, R., 2010. Retiree health insurance and disengagement from a career job. Journal of Labor Research 31 (3), 247–262.

Sociale Verzekeringsbank, 2019. https://www.svb.nl/int/nl/aow/.

Stichting Pensioenfonds Zorg en Welzijn, 2007. Statuten en reglementen.

- Van Duijn, M., Mastrogiacomo, M., Lindeboom, M., Lundborg, P., 2013. Expected and actual replacement rates in the pension system of the Netherlands: how and why do they differ? Journal of Pension Economics and Finance 12 (2), 168–189.
- Vermeer, N., Mastrogiacomo, M., van Soest, A., 2016. Demanding occupations and the retirement age. Labour Economics 43, 159–170.

Appendix

Details of income taxation

After-tax income and health insurance premium payments are calculated according to the rules of the Dutch Tax and Customs Administration (Geppaart, 2010). First, the basis for wage tax and national insurance premiums is determined. Gross income is first determined as the sum of work income, occupational pension benefit, and state pension benefit, all on an annual basis, including holiday allowance and end-of-year bonus.³ From this gross income, the premiums paid to the occupational pension scheme are deducted, since these premiums are tax-deductible.⁴ That gives the basis for the employee insurance premiums include the "lifecycle savings scheme" premiums and private car use.⁵ This yields the basis for income-related health insurance premiums. Finally, the income-related health insurance premiums because the employer fully compensates the employee for this premium, which is therefore treated as taxable income. This gives the basis for income tax and national insurance contributions.

Given the basis for income tax and national insurance premiums, the amount of income tax and social premiums payable is determined according to the progressive tax rates. The tax rates in the first two tax brackets decrease by a certain number of percentage points after the statutory retirement age, because old-age pension insurance contributions no longer apply after this age. The average tax rate increases with income due to the increase in marginal income tax rates over the brackets. Hence, retirees with low income have the lowest income tax in terms of their income.

Finally, tax credits should be accounted for to determine the ultimate amount of tax payable. Some tax credits provide a flat-rate amount, while others are income-related or depend on the domestic situation. The amount of a tax credit can never exceed the amount of tax to be paid (before subtracting the credit) (Belastingdienst, 2019). The tax credits considered in this study are the general tax credit, employed person's tax credit, employed person's tax credit reduction, elderly person's tax credit, and the elderly single person's tax credit. We do not consider the

- 3 For income tax purposes there are three types of taxable income: income from employment (current or past) and home ownership, income from a substantial interest, and income from savings and investments. We consider only the first category.
- 4 Additions to and deductions from gross income include the tax-deductible company savings scheme premiums, wage payments in kind, and claims for future income. In our analysis these are assumed to be zero.
- 5 In our analysis these are assumed to be zero.

work bonus since it was abolished in 2018. We also do not consider the single parent's tax credit, single parent's supplementary tax credit, and the combination tax credit. These tax credits depend on whether people have children living in the household, but relevant data are not available for the subjects of our study.

Given the pension premiums, taxes, tax credits, health insurance premiums, and the health insurance premium compensation, net work income and retirement income can be calculated by adding the tax credits and health insurance premium compensation and subtracting pension premiums, taxes, and health insurance premiums from gross income.

OVERZICHT UITGAVEN IN DE DESIGN PAPER SERIE

- 1 Naar een nieuw pensioencontract (2011) Lans Bovenberg en Casper van Ewijk
- Langlevenrisico in collectieve pensioencontracten (2011)
 Anja De Waegenaere, Alexander Paulis en Job Stigter
- Bouwstenen voor nieuwe pensioencontracten en uitdagingen voor het toezicht daarop (2011)
 Theo Nijman en Lans Bovenberg
- 4 European supervision of pension funds: purpose, scope and design (2011) Niels Kortleve, Wilfried Mulder and Antoon Pelsser
- Regulating pensions: Why the European Union matters (2011)
 Ton van den Brink, Hans van Meerten and Sybe de Vries
- 6 The design of European supervision of pension funds (2012)
 Dirk Broeders, Niels Kortleve, Antoon Pelsser and Jan-Willem Wijckmans
- 7 Hoe gevoelig is de uittredeleeftijd voor veranderingen in het pensioenstelsel? (2012) Didier Fouarge, Andries de Grip en Raymond Montizaan
- 8 De inkomensverdeling en levensverwachting van ouderen (2012) Marike Knoef, Rob Alessie en Adriaan Kalwij
- 9 Marktconsistente waardering van zachte pensioenrechten (2012) Theo Nijman en Bas Werker
- 10 De RAM in het nieuwe pensioenakkoord (2012) Frank de Jong en Peter Schotman
- The longevity risk of the Dutch Actuarial Association's projection model (2012)
 Frederik Peters, Wilma Nusselder and Johan Mackenbach

- 12 Het koppelen van pensioenleeftijd en pensioenaanspraken aan de levensverwachting (2012)
 - Anja De Waegenaere, Bertrand Melenberg en Tim Boonen
- 13 Impliciete en expliciete leeftijdsdifferentiatie in pensioencontracten (2013)
 Roel Mehlkopf, Jan Bonenkamp, Casper van Ewijk, Harry ter Rele en Ed Westerhout
- 14 Hoofdlijnen Pensioenakkoord, juridisch begrepen (2013)
 Mark Heemskerk, Bas de Jong en René Maatman
- 15 Different people, different choices: The influence of visual stimuli in communication on pension choice (2013)
 Elisabeth Brüggen, Ingrid Rohde and Mijke van den Broeke
- 16 Herverdeling door pensioenregelingen (2013)
 Jan Bonenkamp, Wilma Nusselder, Johan
 Mackenbach, Frederik Peters en Harry ter
 Rele
- 17 Guarantees and habit formation in pension schemes: A critical analysis of the floorleverage rule (2013)
 Frank de Jong and Yang Zhou
- 18 The holistic balance sheet as a building block in pension fund supervision (2013) Erwin Fransen, Niels Kortleve, Hans Schumacher, Hans Staring and Jan-Willem Wijckmans
- 19 Collective pension schemes and individual choice (2013)
 Jules van Binsbergen, Dirk Broeders, Myrthe de Jong and Ralph Koijen
- 20 Building a distribution builder: Design considerations for financial investment and pension decisions (2013)
 Bas Donkers, Carlos Lourenço, Daniel Goldstein and Benedict Dellaert

- 21 Escalerende garantietoezeggingen: een alternatief voor het StAr RAM-contract (2013) Servaas van Bilsen, Roger Laeven en Theo Nijman
- 22 A reporting standard for defined contribution pension plans (2013) Kees de Vaan, Daniele Fano, Herialt Mens and Giovanna Nicodano
- 23 Op naar actieve pensioenconsumenten: Inhoudelijke kenmerken en randvoorwaarden van effectieve pensioencommunicatie (2013) Niels Kortleve, Guido Verbaal en Charlotte Kuiper
- 24 Naar een nieuw deelnemergericht UPO (2013) Charlotte Kuiper, Arthur van Soest en Cees Dert
- 25 Measuring retirement savings adequacy; developing a multi-pillar approach in the Netherlands (2013) Marike Knoef, Jim Been, Rob Alessie, Koen Caminada, Kees Goudswaard, and Adriaan Kalwij
- 26 Illiquiditeit voor pensioenfondsen en verzekeraars: Rendement versus risico (2014) Joost Driessen
- 27 De doorsneesystematiek in aanvullende pensioenregelingen: effecten, alternatieven en transitiepaden (2014)
 Jan Bonenkamp, Ryanne Cox en Marcel Lever
- 28 EIOPA: bevoegdheden en rechtsbescherming (2014)
 Ivor Witte

Ivor Witte

- 29 Een institutionele beleggersblik op de Nederlandse woningmarkt (2013) Dirk Brounen en Ronald Mahieu
- 30 Verzekeraar en het reële pensioencontract (2014)
 Jolanda van den Brink, Erik Lutjens en Ivor Witte
- Pensioen, consumptiebehoeften en ouderenzorg (2014)
 Marike Knoef, Arjen Hussem, Arjan Soede en Jochem de Bresser
- 32 Habit formation: implications for pension plans (2014)Frank de Jong and Yang Zhou

- 33 Het Algemeen pensioenfonds en de taakafbakening (2014)Ivor Witte
- 34 Intergenerational Risk Trading (2014) Jiajia Cui and Eduard Ponds
- 35 Beëindiging van de doorsneesystematiek:
 juridisch navigeren naar alternatieven (2015)
 Dick Boeijen, Mark Heemskerk en
 René Maatman
- 36 Purchasing an annuity: now or later? The role of interest rates (2015)
 Thijs Markwat, Roderick Molenaar and Juan Carlos Rodriguez
- 37 Entrepreneurs without wealth? An overview of their portfolio using different data sources for the Netherlands (2015)
 Mauro Mastrogiacomo, Yue Li and Rik Dillingh
- 38 The psychology and economics of reverse mortgage attitudes. Evidence from the Netherlands (2015)
 Rik Dillingh, Henriëtte Prast, Mariacristina Rossi and Cesira Urzì Brancati
- 39 Keuzevrijheid in de uittreedleeftijd (2015) Arthur van Soest
- 40 Afschaffing doorsneesystematiek: verkenning van varianten (2015) Jan Bonenkamp en Marcel Lever
- 41 Nederlandse pensioenopbouw in internationaal perspectief (2015)
 Marike Knoef, Kees Goudswaard, Jim Been en Koen Caminada
- 42 Intergenerationele risicodeling in collectieve en individuele pensioencontracten (2015) Jan Bonenkamp, Peter Broer en Ed Westerhout
- 43 Inflation Experiences of Retirees (2015) Adriaan Kalwij, Rob Alessie, Jonathan Gardner and Ashik Anwar Ali
- 44 Financial fairness and conditional indexation (2015) Torsten Kleinow and Hans Schumacher
- 45 Lessons from the Swedish occupational pension system (2015)
 Lans Bovenberg, Ryanne Cox and Stefan Lundbergh

46 Heldere en harde pensioenrechten onder een PPR (2016) Mark Heemskerk, René Maatman en Bas

Werker

47 Segmentation of pension plan participants: Identifying dimensions of heterogeneity (2016)
Wiebke Eberhardt, Elisabeth Brüggen,

Thomas Post and Chantal Hoet

- 48 How do people spend their time before and after retirement? (2016) Johannes Binswanger
- 49 Naar een nieuwe aanpak voor risicoprofielmeting voor deelnemers in pensioenregelingen (2016) Benedict Dellaert, Bas Donkers, Marc Turlings, Tom Steenkamp en Ed Vermeulen
- 50 Individueel defined contribution in de uitkeringsfase (2016) Tom Steenkamp
- 51 Wat vinden en verwachten Nederlanders van het pensioen? (2016) Arthur van Soest
- 52 Do life expectancy projections need to account for the impact of smoking? (2016) Frederik Peters, Johan Mackenbach en Wilma Nusselder
- 53 Effecten van gelaagdheid in pensioendocumenten: een gebruikersstudie (2016) Louise Nell, Leo Lentz en Henk Pander Maat
- 54 Term Structures with Converging Forward Rates (2016) Michel Vellekoop and Jan de Kort
- 55 Participation and choice in funded pension plans (2016)
 - Manuel García-Huitrón and Eduard Ponds
- 56 Interest rate models for pension and insurance regulation (2016) Dirk Broeders, Frank de Jong and Peter Schotman
- 57 An evaluation of the nFTK (2016) Lei Shu, Bertrand Melenberg and Hans Schumacher
- 58 Pensioenen en inkomensongelijkheid onder ouderen in Europa (2016) Koen Caminada, Kees Goudswaard, Jim Been en Marike Knoef

- 59 Towards a practical and scientifically sound tool for measuring time and risk preferences in pension savings decisions (2016) Jan Potters, Arno Riedl and Paul Smeets
- 60 Save more or retire later? Retirement planning heterogeneity and perceptions of savings adequacy and income constraints (2016)

Ron van Schie, Benedict Dellaert and Bas Donkers

61 Uitstroom van oudere werknemers bij overheid en onderwijs. Selectie uit de poort (2016)

Frank Cörvers en Janneke Wilschut

62 Pension risk preferences. A personalized elicitation method and its impact on asset allocation (2016)

Gosse Alserda, Benedict Dellaert, Laurens Swinkels and Fieke van der Lecq

- 63 Market-consistent valuation of pension liabilities (2016)
 Antoon Pelsser, Ahmad Salahnejhad and Ramon van den Akker
- 64 Will we repay our debts before retirement? Or did we already, but nobody noticed? (2016)

Mauro Mastrogiacomo

- 65 Effectieve ondersteuning van zelfmanagement voor de consument (2016) Peter Lapperre, Alwin Oerlemans en Benedict Dellaert
- 66 Risk sharing rules for longevity risk: impact and wealth transfers (2017) Anja De Waegenaere, Bertrand Melenberg and Thijs Markwat
- 67 Heterogeniteit in doorsneeproblematiek.
 Hoe pakt de transitie naar degressieve opbouw uit voor verschillende pensioenfondsen? (2017)
 Loes Frehen, Wouter van Wel, Casper van Ewijk, Johan Bonekamp, Joost van
 Valkengoed en Dick Boeijen
- 68 De toereikendheid van pensioenopbouw na de crisis en pensioenhervormingen (2017) Marike Knoef, Jim Been, Koen Caminada, Kees Goudswaard en Jason Rhuggenaath

- 69 De combinatie van betaald en onbetaald werk in de jaren voor pensioen (2017) Marleen Damman en Hanna van Solinge
- 70 Default life-cycles for retirement savings (2017)
 Anna Grebenchtchikova, Roderick Molenaar,

Peter Schotman en Bas Werker

 71 Welke keuzemogelijkheden zijn wenselijk vanuit het perspectief van de deelnemer? (2017)
 Casper van Ewijk, Roel Mehlkopf, Sara van

den Bleeken en Chantal Hoet

- 72 Activating pension plan participants:
 investment and assurance frames (2017)
 Wiebke Eberhardt, Elisabeth Brüggen,
 Thomas Post en Chantal Hoet
- 73 Zerotopia bounded and unbounded pension adventures (2017) Samuel Sender
- 74 Keuzemogelijkheden en maatwerk binnen pensioenregelingen (2017)
 Saskia Bakels, Agnes Joseph, Niels Kortleve en Theo Nijman
- 75 Polderen over het pensioenstelsel. Het debat tussen de sociale partners en de overheid over de oudedagvoorzieningen in Nederland, 1945-2000 (2017)
 Paul Brusse
- 76 Van uitkeringsovereenkomst naar PPR (2017) Mark Heemskerk, Kees Kamminga, René Maatman en Bas Werker
- 77 Pensioenresultaat bij degressieve opbouw en progressieve premie (2017) Marcel Lever en Sander Muns
- 78 Bestedingsbehoeften bij een afnemende gezondheid na pensionering (2017) Lieke Kools en Marike Knoef
- 79 Model Risk in the Pricing of Reverse
 Mortgage Products (2017)
 Anja De Waegenaere, Bertrand Melenberg,
 Hans Schumacher, Lei Shu and Lieke Werner
- 80 Expected Shortfall voor toezicht op verzekeraars: is het relevant? (2017) Tim Boonen
- 81 The Effect of the Assumed Interest Rate and Smoothing on Variable Annuities (2017) Anne G. Balter and Bas J.M. Werker

- 82 Consumer acceptance of online pension investment advice (2017)
 Benedict Dellaert, Bas Donkers and Carlos Lourenço
- 83 Individualized life-cycle investing (2017) Gréta Oleár, Frank de Jong and Ingmar Minderhoud
- 84 The value and risk of intergenerational risk sharing (2017) Bas Werker
- 85 Pensioenwensen voor en na de crisis (2017) Jochem de Bresser, Marike Knoef en Lieke Kools
- 86 Welke vaste dalingen en welk beleggingsbeleid passen bij gewenste uitkeringsprofielen in verbeterde premieregelingen? (2017) Johan Bonekamp, Lans Bovenberg, Theo Nijman en Bas Werker
- 87 Inkomens- en vermogensafhankelijke eigen bijdragen in de langdurige ouderenzorg: een levensloopperspectief (2017) Arjen Hussem, Harry ter Rele en Bram Wouterse
- 88 Creating good choice environments Insights from research and industry practice (2017) Elisabeth Brüggen, Thomas Post and Kimberley van der Heijden
- 89 Two decades of working beyond age 65 in the Netherlands. Health trends and changes in socio-economic and work factors to determine the feasibility of extending working lives beyond age 65 (2017) Dorly Deeg, Maaike van der Noordt and Suzan van der Pas
- 90 Cardiovascular disease in older workers. How can workforce participation be maintained in light of changes over time in determinants of cardiovascular disease? (2017) Dorly Deeg, E. Burgers and Maaike van der Noordt
- 91 Zicht op zzp-pensioen (2017) Wim Zwinkels, Marike Knoef, Jim Been, Koen Caminada en Kees Goudswaard
- 92 Return, risk, and the preferred mix of PAYG and funded pensions (2017) Marcel Lever, Thomas Michielsen and Sander Muns

- 93 Life events and participant engagement in pension plans (2017)
 Matthew Blakstad, Elisabeth Brüggen and Thomas Post
- 94 Parttime pensioneren en de arbeidsparticipatie (2017) Raymond Montizaan
- 95 Keuzevrijheid in pensioen: ons brein wil niet kiezen, maar wel gekozen hebben (2018)Walter Limpens en Joyce Vonken
- 96 Employability after age 65? Trends over 23 years in life expectancy in good and in poor physical and cognitive health of 65-74-year-olds in the Netherlands (2018) Dorly Deeg, Maaike van der Noordt, Emiel Hoogendijk, Hannie Comijs and Martijn Huisman
- 97 Loslaten van de verplichte pensioenleeftijd en het organisatieklimaat rondom langer doorwerken (2018) Jaap Oude Mulders, Kène Henkens en Harry van Dalen
- 98 Overgangseffecten bij introductie degressieve opbouw (2018) Bas Werker
- 99 You're invited RSVP! The role of tailoring in incentivising people to delve into their pension situation (2018)
 Milena Dinkova, Sanne Elling, Adriaan Kalwij en Leo Lentz
- 100 Geleidelijke uittreding en de rol van deeltijdpensioen (2018)Jonneke Bolhaar en Daniël van Vuuren
- 101 Naar een model voor pensioencommunicatie (2018)
 Leo Lentz, Louise Nell en Henk Pander Maat
- 102 Tien jaar UPO. Een terugblik en vooruitblik op inhoud, doelen en effectiviteit (2018) Sanne Elling en Leo Lentz
- Health and household expenditures (2018)
 Raun van Ooijen, Jochem de Bresser en Marike Knoef
- 104 Keuzevrijheid in de uitkeringsfase: internationale ervaringen (2018)
 Marcel Lever, Eduard Ponds, Rik Dillingh en Ralph Stevens

- 105 The move towards riskier pension products in the world's best pension systems (2018) Anne G. Balter, Malene Kallestrup-Lamb and Jesper Rangvid
- 106 Life Cycle Option Value: The value of consumer flexibility in planning for retirement (2018)
 Sonja Wendel, Benedict Dellaert and Bas Donkers
- 107 Naar een duidelijk eigendomsbegrip (2018) Jop Tangelder
- 108 Effect van stijging AOW-leeftijd op arbeidsongeschiktheid (2018)
 Rik Dillingh, Jonneke Bolhaar, Marcel Lever, Harry ter Rele, Lisette Swart en Koen van der Ven
- 109 Is de toekomst gearriveerd? Data science en individuele keuzemogelijkheden in pensioen (2018)
 Wesley Kaufmann, Bastiaan Starink en Bas Werker
- 110 De woontevredenheid van ouderen in Nederland (2018) Jan Rouwendal
- 111 Towards better prediction of individual longevity (2018)
 Dorly Deeg, Jan Kardaun, Maaike van der Noordt, Emiel Hoogendijk en Natasja van Schoor
- 112 Framing in pensioenkeuzes. Het effect van framing in de keuze voor beleggingsprofiel in DC-plannen naar aanleiding van de Wet verbeterde premieregeling (2018) Marijke van Putten, Rogier Potter van Loon, Marc Turlings en Eric van Dijk
- 113 Working life expectancy in good and poor self-perceived health among Dutch workers aged 55-65 years with a chronic disease over the period 1992-2016 (2019) Astrid de Wind, Maaike van der Noordt, Dorly Deeg and Cécile Boot
- 114 Working conditions in post-retirement jobs: A European comparison (2019)Ellen Dingemans and Kène Henkens

- Is additional indebtedness the way to increase mortgage-default insurance coverage? (2019)
 Yeorim Kim, Mauro Mastrogiacomo, Stefan Hochguertel and Hans Bloemen
- Appreciated but complicated pension
 Choices? Insights from the Swedish
 Premium Pension System (2019)
 Monika Böhnke, Elisabeth Brüggen and
 Thomas Post
- 117 Towards integrated personal financial planning. Information barriers and design propositions (2019)
 Nitesh Bharosa and Marijn Janssen
- 118 The effect of tailoring pension information on navigation behavior (2019)
 Milena Dinkova, Sanne Elling, Adriaan
 Kalwij and Leo Lentz
- 119 Opleiding, levensverwachting en pensioenleeftijd: een vergelijking van Nederland met andere Europese landen (2019)
 Johan Mackenbach, José Bubio Valverda

Johan Mackenbach, José Rubio Valverde en Wilma Nusselder

- Giving with a warm hand: Evidence on estate planning and bequests (2019)
 Eduard Suari-Andreu, Raun van Ooijen, Rob J.M. Alessie and Viola Angelini
- 121 Investeren in menselijk kapitaal: een gecombineerd werknemers- en werkgeversperspectief (2019) Raymond Montizaan, Merlin Nieste en Davey Poulissen
- 122 The rise in life expectancy corresponding rise in subjective life expectancy? Changes over the period 1999–2016 (2019)
 Dorly Deeg, Maaike van der Noordt, Noëlle Sant, Henrike Galenkamp, Fanny Janssen and Martijn Huisman
- 123 Pensioenaanvullingen uit het eigen woningbezit (2019)Dirk Brounen, Niels Kortleve en Eduard Ponds
- Personal and work-related predictors of early exit from paid work among older workers with health limitations (2019)
 Nils Plomp, Sascha de Breij and Dorly Deeg

- 125 Het delen van langlevenrisico (2019)Anja De Waegenaere, Agnes Joseph, Pascal Janssen en Michel Vellekoop
- 126 Maatwerk in pensioencommunicatie (2019)S.K. Elling en L.R. Lentz
- 127 Dutch Employers' Responses to an Aging Workforce: Evidence from Surveys, 2009– 2017 (2019)
 Jaap Oude Mulders, Kène Henkens and Hendrik P. van Dalen
- Preferences for solidarity and attitudes towards the Dutch pension system – Evidence from a representative sample (2019)
 Arno Riedl, Hans Schmeets and Peter Werner
- 129 Deeltijdpensioen geen wondermiddel voor langer doorwerken (2019)
 Henk-Wim de Boer, Tunga Kantarcı,
 Daniel van Vuuren en Ed Westerhout
- 130 Spaarmotieven en consumptiegedrag (2019) Johan Bonekamp en Arthur van Soest
- 131 Substitute services: a barrier to controlling long-term care expenditures (2019) Mark Kattenberg and Pieter Bakx
- 132 Voorstel keuzearchitectuur pensioensparen voor zelfstandigen (2019) Jona Linde
- The impact of the virtual integration of assets on pension risk preferences of individuals (2019)
 Sesil Lim, Bas Donkers en Benedict Dellaert
- 134 Reforming the statutory retirement age:
 Policy preferences of employers (2019)
 Hendrik P. van Dalen, Kène Henkens and
 Jaap Oude Mulders
- 135 Compensatie bij afschaffing doorsneesystematiek (2019)
 Dick Boeijen, Chantal de Groot, Mark Heemskerk, Niels Kortleve en René Maatman
- 136 Debt affordability after retirement, interest rate shocks and voluntary repayments
 (2019)
 Mauro Mastrogiacomo

- 137 Using social norms to activate pension plan members: insights from practice (2019)
 Joyce Augustus-Vonken, Pieter Verhallen,
 Lisa Brüggen and Thomas Post
- 138 Alternatieven voor de huidige verplichtstelling van bedrijfstakpensioenfondsen (2020)
 Erik Lutiens en Fieke van der Lecq
- 139 Eigen bijdrage aan ouderenzorg (2020) Pieter Bakx, Judith Bom, Marianne Tenand en Bram Wouterse
- 140 Inrichting fiscaal kader bij afschaffing doorsneesystematiek (2020)Bastiaan Starink en Michael Visser
- 141 Hervorming langdurige zorg: trends in het gebruik van verpleging en verzorging (2020)
 Pieter Bakx, Pilar Garcia-Gomez, Sara Rellstab, Erik Schut en Eddy van Doorslaer
- 142 Genetic health risks, insurance, and retirement (2020)Richard Karlsson Linnér and PhilippD. Koellinger
- 143 Publieke middelen voor particuliere ouderenzorg (2020)Arjen Hussem, Marianne Tenand en Pieter Bakx
- 144 Emotions and technology in pension service interactions: Taking stock and moving forward (2020)
 Wiebke Eberhardt, Alexander Henkel en Chantal Hoet
- 145 Opleidingsverschillen in levensverwachting: de bijdrage van acht risicofactoren (2020)
 Wilma J. Nusselder, José Rubio Valverde en Johan P. Mackenbach
- 146 Shades of Labor: Motives of Older Adults to Participate in Productive Activities (2020) Sonja Wendel and Benedict Dellaert
- 147 Raising pension awareness through letters and social media: Evidence from a randomized and a quasi-experiment (2020) Marike Knoef, Jim Been and Marijke van Putten

- 148 Infographics and Financial Decisions (2020) Ruben Cox and Peter de Goeij
- 149 To what extent can partial retirement ensure retirement income adequacy? (2020)Tunga Kantarcı and Jochem Zweerink



Network for Studies on Pensions, Aging and Retirement

This is a publication of: Netspar Phone +31 13 466 2109 E-mail info@netspar.nl www.netspar.nl

May 2020